Author's response to reviews

Title: An equivalence evaluation of a nurse-moderated group-based internet support program for new mothers versus standard care: A pragmatic preference randomised controlled trial.

Authors:

Alyssa C P Sawyer (alyssa.sawyer@adelaide.edu.au)
John Lynch (john.lynch@adelaide.edu.au)
Kerrie Bowering (kerrie.Bowering@health.sa.gov.au)
Debra Jeffs (debra.Jeffs@health.sa.gov.au)
Jenny Clark (jenny.Clark@health.sa.gov.au)
Christine Mpundu-Kaambwa (christine.mpundu-kaambwa@adelaide.edu)
Michael G Sawyer (michael.sawyer@adelaide.edu.au)

Version: 2
Date: 24 April 2014

Author's response to reviews: see over
24 April 2014

Dear Editor,


This paper is a protocol describing a randomised controlled trial that we are currently conducting. Recruitment to the trial was completed in December 2013, and delivery of the intervention is ongoing.

The trial has received funding from the National Health and Medical Research Council and ethics approval from Children, Youth & Women’s Health Service Human Research Ethics Committee. As requested via email on the 23 April 2014 we have removed the documentary proof of ethical approval and peer-reviewed funding from this letter.

Currently we have not published any of the results from the trial. This paper has not been published or considered for publication elsewhere.

Thank you for your consideration of this paper. We look forward to hearing from you.

Yours sincerely,

Dr Alyssa Sawyer